

Original citation:

Anderson, Roy M., Truscott, James E., Pullan, Rachel L., Brooker, Simon J. and Hollingsworth, T. Déirdre. (2013) How effective is school-based deworming for the community-wide control of soil-transmitted helminths? PLoS Neglected Tropical Diseases, Vol.7 (No.2). Article no. e2027.

Permanent WRAP url:

http://wrap.warwick.ac.uk/53261

Copyright and reuse:

The Warwick Research Archive Portal (WRAP) makes the work of researchers of the University of Warwick available open access under the following conditions.

This article is made available under the Creative Commons Attribution-NonCommercial-NoDerivs 3.0 Unported (CC BY-NC-ND 3.0) license and may be reused according to the conditions of the license. For more details see: <u>http://creativecommons.org/licenses/bync-nd/3.0/</u>

A note on versions:

The version presented in WRAP is the published version, or, version of record, and may be cited as it appears here.

For more information, please contact the WRAP Team at: wrap@warwick.ac.uk

warwick**publications**wrap

highlight your research

http://go.warwick.ac.uk/lib-publications

How Effective Is School-Based Deworming for the Community-Wide Control of Soil-Transmitted Helminths?

Roy M. Anderson¹*, James E. Truscott², Rachel L. Pullan³, Simon J. Brooker^{3,4}, T. Deirdre Hollingsworth⁵*

1 London Centre for Neglected Tropical Diseases, Department of Infectious Disease Epidemiology, School of Public Health, Imperial College London, London, United Kingdom, 2 Partnership for Child Development, Department of Infectious Disease Epidemiology, School of Public Health, Imperial College London, London, United Kingdom, 3 Faculty of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom, 4 Kenya Medical Research Institute–Wellcome Trust Research Programme, Nairobi, Kenya, 5 MRC Centre for Outbreak Analysis and Modelling, Department of Infectious Disease Epidemiology, School of Public Health, Imperial College London, London, United Kingdom

Abstract

Background: The London Declaration on neglected tropical diseases was based in part on a new World Health Organization roadmap to "sustain, expand and extend drug access programmes to ensure the necessary supply of drugs and other interventions to help control by 2020". Large drug donations from the pharmaceutical industry form the backbone to this aim, especially for soil-transmitted helminths (STHs) raising the question of how best to use these resources. Deworming for STHs is often targeted at school children because they are at greatest risk of morbidity and because it is remarkably cost-effective. However, the impact of school-based deworming on transmission in the wider community remains unclear.

Methods: We first estimate the proportion of parasites targeted by school-based deworming using demography, school enrolment, and data from a small number of example settings where age-specific intensity of infection (either worms or eggs) has been measured for all ages. We also use transmission models to investigate the potential impact of this coverage on transmission for different mixing scenarios.

Principal Findings: In the example settings <30% of the population are 5 to <15 years old. Combining this demography with the infection age-intensity profile we estimate that in one setting school children output as little as 15% of hookworm eggs, whereas in another setting they harbour up to 50% of *Ascaris lumbricoides* worms (the highest proportion of parasites for our examples). In addition, it is estimated that from 40–70% of these children are enrolled at school.

Conclusions: These estimates suggest that, whilst school-based programmes have many important benefits, the proportion of infective stages targeted by school-based deworming may be limited, particularly where hookworm predominates. We discuss the consequences for transmission for a range of scenarios, including when infective stages deposited by children are more likely to contribute to transmission than those from adults.

Citation: Anderson RM, Truscott JE, Pullan RL, Brooker SJ, Hollingsworth TD (2013) How Effective Is School-Based Deworming for the Community-Wide Control of Soil-Transmitted Helminths? PLoS Negl Trop Dis 7(2): e2027. doi:10.1371/journal.pntd.0002027

Editor: Patrick J. Lammie, Centers for Disease Control and Prevention, United States of America

Received June 1, 2012; Accepted December 7, 2012; Published February 28, 2013

Copyright: © 2013 Anderson et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The authors gratefully acknowledge grant support for this research from the Bill and Melinda Gates Foundation and the Partnership for Child Development. SJB is supported by a Senior Fellowship in Basic Biomedical Science from the Wellcome Trust (098045), which also funds RLP. RMA thanks GSK for Centre Funding. TDH gratefully acknowledges Imperial College London for funding of her Junior Research Fellowship, the Vaccine Modelling Initiative, funded by the Bill and Melinda Gates Foundation, and the Medical Research Council for Centre Funding. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: Roy M. Anderson is non-executive member of the board of GlaxoSmithKline (GSK). GlaxoSmithKline provided no financial support for the work and played no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

* E-mail: roy.anderson@imperial.ac.uk (RMA); d.hollingsworth@imperial.ac.uk (TDH)

Introduction

In January 2012, a high-level meeting brought together 13 pharmaceutical companies and the global health community in London, UK to announce a new public-private partnership to eliminate or control the seven preventable neglected tropical diseases (NTDs) by 2020, based largely on a new NTD roadmap from the World Health Organization (WHO) [1]. The pledge by pharmaceutical companies to sustain and extend donation programmes facilitates a large portion of the necessary supply of drugs and other interventions to help achieve this goal [2]. Such commitments raise the question of how best to use these resources to induce maximum impact, given that many treatments for NTDs

must be administered repeatedly to individuals living in endemic areas due to re-exposure to infection and the absence of fully protective acquired immunity.

The most common NTDs worldwide are the soil-transmitted helminths (STH: Ascaris lumbricoides, Trichuris trichiura and the hookworms, Necator americanus and Ancylostoma duodenale), with an estimated 5.3 billion people worldwide, including 1.0 billion schoolaged children, living in areas of stable transmission for at least one STH species [3]. STHs are easily treated with one of four drugs: albendazole and mebendazole, and to a lesser extent, levamisole, and pyrantel pamoate [4–5]. However, reinfection commonly occurs [6] due to the inability of the human host to mount protective immunity to reinfection by intestinal helminths [7–8], combined

Author Summary

Large donations of drugs to treat soil-transmitted helminths (STHs, intestinal worms) means that many more school-aged children will be treated, improving their wellbeing and development. These children will have to be repeatedly treated since reinfection will occur due to contaminated environments in the absence of improvements in hygiene and sanitation. Repeated treatment of school-aged children may have the added benefit of reductions in levels of infection for the whole community. This will in part be determined by the proportion of the total worms harboured or eggs output by school-aged children, a product of how heavily infected school-aged children are and how many school-aged children there are in the community. In one setting school-aged children output as little as 15% of hookworm eggs whereas in another setting they harbour up to 50% of roundworms. Thus, whilst school-based programmes may have important health benefits, the community-level impact on transmission could be limited unless school-aged children over-contribute to infection. We use mathematical models to show that if children contribute more infective stages to the environment which adults are exposed to than adults do, the reductions in transmission resulting from treating children will be larger, but may still be limited.

with inadequate hygiene and sanitation to restrict or eliminate reexposure in environments continuously contaminated with the egg or larval free-living transmission stages of these parasitic worms [9].

Following treatment, average worm loads in the population return to their pre-treatment equilibiria in a monotonic manner. The exact dynamics will depend on a number of densitydependent processes that influence parasite reproduction, infection and mortality (in part related to the build-up of a degree of acquired immunity), plus the relatively long life expectancies of established worms in the human host (measured in years) [10]. It will also depend on the proportion of worms in the entire human community in a defined location which are exposed to treatment in a particular control programme.

Deworming programmes for the STHs are often centred on school delivery because of the large burden of morbidity and concomitant developmental consequences for these children [11-13], as well as relative ease of access to children in poor rural areas through schools and the cost-effectiveness of school-based deworming [14-15]. A number of countries have programmes which additionally include adults, as part of lymphatic filariasis treatment campaigns providing mass treatment with albendazole and ivermectin (or diethylcarbamazine), with associated large impacts on transmission [16]. However, there are large areas where this STHs are not co-endemic for lymphatic filariasis and these areas the WHO-recommended treatment strategies prioritise school-aged children, but also recommend preventive chemotherapy of preschool children, women of childbearing age and adults at high risk [4]. A number of countries are currently implementing only school-based deworming [11]. The large donation of 600 million doses per year announced in the London Declaration almost completely covers the estimated 610 million school-aged children in need of preventive chemotherapy [1-2,11,17], but it does not cover pre-school children, women of child-bearing age or treatments more than once a year. This paper therefore examines the effectiveness of school-targeted programmes in restricting transmission within the larger community of pre-school children, school-aged children and adults. We use analytical methods deriving from the description of the transmission dynamics of these parasitic worms [18] and demographic plus school attendance information to calculate what fraction of the total population is treated. We also use mathematical models to discuss the impact of school-based programmes on transmission, including scenarios in which infective stages deposited by children are more likely to contribute to transmission than those from adults. We discuss how the impact of a treatment programme could affect the infection dynamics in the population as a whole depending on the, as yet unknown, details of between age-group mixing.

Methods

The effect of treating school-aged children on the overall transmission dynamics of the parasites depends on a number of factors. These include the fraction of the total worm population harboured by these age groups, the fraction of the eligible schoolaged children who attend school and receive treatment, how other age groups are exposed to the eggs or infective larvae produced by school-aged children and vice versa, and drug efficacy. Data exist for the first two factors, which we discuss below. However there is limited evidence to facilitate the exploration of various assumptions such as random or non-random exposure of age groups to the infective stages produced by other age groups in a defined human community. Drug pharmacokinetics and efficacy are also reasonably well documented for the commonly used drugs for STHs and schistosomes (mebendazole [19], albendazole [20-22] and praziquantel [23]). In this analysis we estimate the fraction of worms among school-aged children and then discuss the likely impact of this level of coverage on transmission using a suite of infectious disease models. We first outline the parasitological measures and data on which these estimates will be based.

Epidemiological Measures - Prevalence and Intensity

Monitoring and evaluation of the impact of community-based preventive chemotherapy programmes is based on two epidemiological measures, the prevalence and the intensity of infection. Prevalence represents the fraction or percentage of the population infected and is typically stratified by factors such as age and gender. The intensity of infection or worm burden for STHs is typical measured indirectly by counts of eggs expelled in faeces (eggs per gram of faeces or EPG) and is similarly typically stratified by age and sex. Less commonly, worm burden may be measured by the worms expelled in total faecal output over a defined period post curative chemotherapy [24-26]. The two epidemiological measures are statistics of the probability distribution of worm numbers per person. These distributions are typically highly aggregated in form, with the variance exceeding the mean in value. They are well described by the negative binomial probability model [27]. For this distribution the relationship between prevalence P (as a proportion) and mean intensity M is given by:

$$P = 1 - (1 + M/k)^{-k} \tag{1}$$

Here, the negative binomial parameter k varies inversely with the degree of parasite aggregation within the human population and for values in excess of five the distribution is approximately random in form with the variance approaching the mean in value. Some typical estimates of the magnitude of k are recorded in Table 1.

A plot of the relationship between P M and k for the negative binomial is presented in Figure 1. It is clear from this figure that the prevalence is a very poor measure of the impact of community-based chemotherapy. Large changes in average worm load as a consequence of treatment will only have a small effect on prevalence unless the mean worm burden is low (i.e. when transmission is very **Table 1.** Estimates for the negative binomial parasite aggregation parameter, k.

Parasite	k value or range	Distribution measure	Data source
Ascaris lumbricoides	0.80	Worm numbers	Elkins et al. (1986) [24]
Ascaris lumbricoides	0.81	Worm numbers	Croll et al. (1982) [34]
Ascaris lumbricoides	0.44	Worm numbers	Martin et al. (1983) [58]
Ascaris lumbricoides	0.46	Worm numbers	Thein-Hlaing et al. (1987) [33]
Ascaris lumbricoides	0.36–0.54	Worm numbers	Chai et al. (1985) [59]
Ascaris lumbricoides	0.59	Worm numbers	Bundy et al. (1987) [35]
Necator americanus	0.34	Eggs per gram of faeces	Bradley et al. (1993) [60]
Necator americanus	0.33–0.61	Eggs per gram of faeces	Quinnell et al. (1993) [36,61]
Trichuris trichiura	0.11-0.65	Eggs per gram of faeces	Bundy et al. (1987) [35]

Small k indicates higher parasite aggregation.

doi:10.1371/journal.pntd.0002027.t001

low). For example, in a study in Myanmar two villages with mean EPGs of about 4000 and about 400, an order of magnitude difference in intensity, had almost no difference in prevalence (Figure 1B [28]) and so highly effective treatment of the high intensity village, reducing the burden by a factor of 10, might be viewed as a failed programme if only prevalence were monitored.

The cost and difficulty of monitoring intensity, as opposed to prevalence, is of course greater. However, if monitoring is to have any value, intensity must be measured in some fraction of the treated population. The issue of how best to sample to gain an accurate picture of the impact of treatment, while attempting to keep sample size and the concomitant costs low, requires careful thought. The underlying distribution of parasite numbers per host is central and given its heterogeneity, small sample sizes will not provide robust measures of trends [29]. One compromise is to monitor impact in a subset of age classes – one of children and one in the adult age groups, to see how the treatment of the schoolaged children impacts on transmission to adults [30].

Surveillance and Epidemiological Databases

The greatly expanded deworming programmes seen in many regions of the world in recent years have not been accompanied by systematic recording of treatments delivered and the associated impact on prevalence and intensity. Recently, however, some progress has been made on the generation of open access databases recording global and national spatial distributions of helminths based on estimates of the infection prevalence, as illustrated, by the Global Atlas of Helminth Infection (http:// www.thiswormyworld.org [12]) and the Global Neglected Tropical Diseases database (http://www.gntd.org [31]). The Global Atlas of Helminth Infection will be expanded in the near future to include measures of the intensity of infection and treatments delivered. The website will also be extended to encourage the electronic deposition of data on STHs collected in association with the current expanded efforts on community-based control using mass or school-aged targeted anthelminthic treatment. Such data are collected by a number of excellent NTD programmes but is rarely subject to detailed analysis on trends in transmission [32].

The present absence of international databases on treatment of STH and impact of such treatment does make analysis of questions concerning the optimal delivery strategy for community-based programme somewhat challenging. We therefore base our analyses on a small number of available well-designed studies that record prevalence and intensity of infection, stratified by age and sex, before and after various treatment programmes. The ageprofiles used are from studies of *A. lumbricoides* in Myanmar [33], India [24] and Iran [34]; *T. trichiura* in St Lucia [35]; hookworm in Uganda [36] and Vietman [37] and, for comparison, studies of *Schistosoma mansoni* in Uganda [38] and Brazil [39].

Fraction of the Total Worm Population in School-Aged Children (5 to 14 Years of Age)

Epidemiological studies that record the mean intensity of infection stratified by age, when combined with demographic plus school enrolment data, provide information on the fraction of the total worm population exposed to treatment. Along with measures of drug efficacy, this in turn gives the fraction of the total worm population removed by school-aged targeted chemotherapy. The importance of this fraction to the overall transmission dynamics of the target parasites cannot be overstated – and in current control programmes it is an unmeasured parameter.

If p(a) is the proportion of the total population in age class a, the proportion of the human population in the school age classes 5 to 14 years of age, p_{5-14} is given by:

$$p_{5-14} = \int_{5}^{15} p(a)da / \int_{0}^{\infty} p(a)da$$
 (2)

The proportion of the total worm population harboured at time t by people between the ages of 5 to 14 years, Q_{5-14} , is then given by:

$$Q_{5-14} = \left[\int_{5}^{15} p(a)M(a)da \right] / \left[\int_{0}^{\infty} p(a)M(a)da \right]$$
(3)

where M(a,t) is the mean worm burden in age-group a at time t. If the measure of intensity of infection is eggs per gram of faeces

(EPG), then the proportion of egg output produced by school-aged children is:

$$epg_{5-14} = \left[\int_{5}^{15} p(a)M(a)f_2(M,k)da \right] / \left[\int_{5}^{\infty} p(a)M(a)f_2(M,k)da \right]$$
(4)

Here f_2 is the density-dependent egg output function, which gives the expected egg output for an individual with mean worm burden M for age a and negative binomial aggregation parameter k. In



Figure 1. The relationship between mean intensity and prevalence. A The relationship between the mean intensity of infection, M, the prevalence of infection, P, and the negative binomial aggregation parameter, k as described by the relationship in equation 1. **B** Relationship between the prevalence and intensity of infection as observed in a study of *A. lumbricoides* [28]. The solid line is the predicted relationship between mean prevalence of infection and worm burden described in equation 1 and plotted in A fitted to estimate the aggregation parameter, k = 0.194. doi:10.1371/journal.pntd.0002027.g001

practice egg output is usually measured, rather than mean worm burden, so we can approximate this by

calculate these fractions for some example datasets on parasite distributions, together with demographic and school enrollment data.

$$epg_{5-14} = \left[\int_{5}^{15} p(a)E(a,t)da \right] / \left[\int_{5}^{\infty} p(a)E(a,t)da \right]$$
(5)

Demographic Data

Demographic data from various countries where STHs are endemic provides an initial template to assess these issues [40].

where E(a,t) the mean egg output in age-group a at time t. We

Table 2. Percentage of population aged 5 to 14 years in 2011 [40].

Country	Percentage of population 5-14 years of age	Total population in millions
Uganda	29.7	33.4
Nigeria	26.9	161.6
Rwanda	26.3	11.1
Kenya	26.3	41.9
Burundi	23.7	8.4
India	19.7	1190.0
Myanmar	18.4	54.0
Thailand	13.4	41.9
People's Republic of China	11.8	1340.0
United Kingdom	11.2	62.7

doi:10.1371/journal.pntd.0002027.t002

Table 2 records the fraction of various populations in the school ages of 5 up to 14 years. In general, within countries where helminth infections are endemic, the fraction of the total population in the school going groups is between 11% and 30%.

School Enrolment Data

UNESCO and the World Bank provide data on school enrolment by sex, location (urban or rural) and country. Recent data are recorded in Figure 2 for rural and urban areas for a selection of countries [41] and in Table 6 for enrollement of female students. Over the past decade there has been a steady increase in school enrolment in most countries throughout the world. Progress has been less good in poor rural areas by comparison with urban districts in developing countries. Generally, the most recent data (2005 and beyond) suggest figures in the 80% to 90% range for most urban areas, but with a range of 20% to 60% in some sub-



Figure 2. School attendance for a selection of countries. This figure was generated by data published by UNICEF for 2005–2010 [41].. For each country there is net attendance rate at primary, in urban (open circles) and rural areas (closed circles) and a net attendance rate for secondary schools (filled squares).

doi:10.1371/journal.pntd.0002027.g002

Table 3. Published estimates of the basic reproductive number R_0 for various helminths.

Parasite	Basic reproductive number, R_0	Measure on which parameter estimated	Data source
Necator americanus	2.0	Re-infection post treatment	Bradley et al. (1993) [60]
Ascaris lumbricoides	4.3	Re-infection post treatment	Croll et al. (1982) [34]
Ascaris lumbricoides	1.7	Re-infection post treatment	Thein-Hlaing et al. (1987) [33]
Trichuris trichiura	4–6	Re-infection post treatment	Bundy et al. (1987) [35]

doi:10.1371/journal.pntd.0002027.t003

Saharan African countries in rural areas. There is often a gender bias in many poor countries, with attendance figures for females lower than those for males in the primary and secondary school enrolment data. Poor attendance could severely reduce the population-level impact of school-based deworming. Conversely however, there is anecdotal evidence that there may be higher attendance to schools for deworming days due to awareness of the health benefits. These effects have not yet been quantified, to our knowledge.

Effect of coverage on Programme Impact

For a given coverage of a school-based programme, as defined by proportion of estimated treatment of a proportion of worms, Q_{5-14} , or egg output, epg_{5-14} , the impact on transmission will depend on the particular dynamics of the parasite. Here we outline general insights on the non-linear effects of limited coverage on transmission. We then use heterogeneous mixing models to investigate how different mixing patterns between adults and children will affect the impact of targeted programmes.

The Basic Reproductive Number (R_0) and Parasite Life Expectancy - Their Effects on the Impact of Community-Based Chemotherapy

Simple theory provides some important general insights into the factors controlling helminth transmission and the impact of community-based chemotherapy [18]. For directly transmitted helminths with a free living larval or egg stage outside the human host, the basic dynamics of the system can be described by the following differential equation,

$$\frac{\mathrm{d}M}{\mathrm{dt}} = (\mu + \mu_1) [R_0 \mathbf{f}(M, z, k) - 1] M \tag{6}$$

Table 4. Published estimates of parasite life expectancy, L_1 , in the human host [18,27].

Parasite	Life expectancy in years, $L_{\rm l}$
Enterobius vermicularis	<1
Trichuris trichiura	1–2
Ascaris lumbricoides	1–2
Necator americanus	2–3
Ancylostoma duodenale	2–3
Schistosoma mansoni	3–5
Schistosoma haematobium	3–5

doi:10.1371/journal.pntd.0002027.t004

where $1/\mu$ is human life expectancy, $1/\mu_1$ is adult parasite life expectancy, M is the mean number of worms in the population, f(M) is the mean egg output per gram of stool, given a mean worm burden of M, the dispersal parameter, k, and fecundity coefficient, z:

$$f(M) = \lambda M \left[1 + \frac{1}{k} M (1 - z) \right]^{-(k+1)}$$
(7)

The basic reproductive number defined as the average number of female worm offspring that survive to reproduce in the absence of density-dependent constraints (ignoring the complexities of mating probabilities and age structure in the human population) is given by equation 8

$$R_0 = s\lambda\beta d_1 d_2 / [(\mu + \mu_1)\mu_2]$$
(8)

Here, s denotes the fraction of female worms, λ the per capita egg production rate, d_1 the proportion of female worms that survive in the human host to reproductive maturity, d_2 the fraction of eggs or larvae that survive to the infective state, $1/\mu$ is human life expectancy, $1/\mu_1$ is the adult worm life expectancy in the human host, $1/\mu_2$ is infective stage life expectancy and β is the per capita transmission coefficient for the infective stage. Table 3 records published estimates of R_0 for various helminth species.

A rough approximation of the growth rate Λ of the parasite population post extensive treatment, again ignoring the effects of density-dependence and worm mating probabilities, is given by:

$$\Lambda \approx (R_0 - 1)/L_1 \tag{9}$$

Where L_1 (=1/ μ_1) is the parasite life expectancy in the human host. Equation 9 is based on the assumption that L_1 is much shorter than human life expectancy, which is true for all STHs (see Table 4).

Ignoring age structure in the human population, the effect of treatment on the effective reproduction number, R_e , is described by

$$R_e = R_0 + \frac{L_1}{\tau} \ln(1 - gh) \tag{10}$$

where τ is the interval between successive rounds of treatment and h is the efficacy of the drug, defined as the proportion of worms killed by the drug. The critical fraction of the human population that must be treated, g_c , to reduce the effective reproductive number R_e to less than unity in value (assuming that the breakpoint in transmission is close to zero due to the aggregated



Figure 3. The proportion of A. lumbricoides worms in children aged 5-14, calculated from equation 3. The demography of the population, A, results in a proportion of 18.4% of the population aged 5-14 years old [45]. Combining this distribution with B the distribution of worms per person by age [57] gives 49.5% of worms in the school-aged group. doi:10.1371/journal.pntd.0002027.g003

85-89 95-99

distribution of worms per person [27]) is given by:

10-14 20-24 30-34

5

0 0-4

$$g_c = (1 - \exp\{(1 - R_0)\tau/L\})/h$$
(11)

40-44 50-54

Age group(years)

60-64

70-74

80+

And thus, the fraction g_c is simply determined by R_0 , τ , h and adult worm life expectancy L_1 . We simulate these approximations for a range of parameter values to illustrate the non-linearities in these relationships.

Heterogeneity Between Age Classes in Contact with Infective Stages

These approximations for homogenous mixing models are very useful to illustrate the impact of limited coverage on transmission. To gain a more accurate picture of how the treatment of just school-aged children influences the overall transmission dynamics of the parasite, the impact of not only age structure but also exposure to infective stages produced by all age groups in the population must be taken into account. The homogeneous model described above can be extended to an age-structured model to describe the characteristic age-profiles of intensity [18]. Such models usually assume that exposure to infective stages across all age classes is random and independent of which age class produces the eggs or infective larvae. However, the spatial structure of egg deposition and infective stage development arising from one age group, plus their concomitant contact with other age groups, is in reality unlikely to be random in a defined community. Instead, it may be more likely that infective stages produced by school-aged children are deposited in areas closer to habitation, and hence acquired by all age groups of the population, while those arising from adults are less likely to come into contact with children. Therefore the model should include heterogeneous mixing [42].

A simple way to mimic non-random contact is to stratify the population into two age groups, namely; school-aged children (5-



Hookworm - eggs per gram 1,600 Uganda - Pullan et al. (2010) 1,400 Mean intensity (EPG) 000 000 000 Percentage of epg in 5-14 year age group - 15.7% 400 200 0 10-14 20-24 30-34 40-44 50-54 0-4 60-64 Age group (years)

в

Figure 4. The proportion of hookworm eggs deposited by children aged 5–14, calculated from equation 5. The demography of the population, A, gives a proportion of 31.2% of the population aged 5–14 years old [45]. Combining this distribution with B the distribution of egg output by age [36] gives 15.7% of worms in the school-aged group. doi:10.1371/journal.pntd.0002027.g004

14 years), and the rest (0–4 and ≥ 15 years combined for simplicity, although patterns may also be different between these two groups), and assume different contact patterns with the infective stages within and between these larger age groupings. Such a stratification of hosts has the further advantage of facilitating the modelling of school-based treatment programmes. We assume that the child and adult age groups have negativebinomially distributed worm distributions with the same shape parameter, k, but different means, M_c and M_a , respectively. The means evolve independently according to the degree of contact of each group with a common infectious 'reservoir'. The model equations are (extending the approach by Chan et al. [42])

$$\frac{dM_c}{dt} = \beta_c l - \sigma M_c \qquad (12)$$

$$\frac{dM_a}{dt} = \beta_a l - \sigma M_a$$

The quantity, l, is the per capita infectiousness of the shared reservoir. The parameters β_c and β_a determine the strength of contact with the reservoir for children and adults respectively. The dynamics of the infectious reservoir are described by the following equation

$$\frac{d}{dt}l = \frac{R_0\mu_2\sigma}{\lambda(\beta_c n_c p + \beta_a n_a q)} [f_2(M_c,k)n_c p + f_2(M_a,k)n_a p] - \mu_2 l \quad (13)$$

The parameters in equation 13 are as defined earlier. The parameters n_c and n_a represent the proportion of the population in the two age classes and p and q the fraction of egg output that enters the reservoir from children aged 5–14 years and other age groups, respectively.

We investigate the effect of regular school-based treatment on the evolution of worm burdens in the community for three scenarios: Table 5. Fraction of worm population or egg output in 5–14 year olds (equations 3 and 5).

	Percentage of the total population			
Parasite	the 5–14 year old age groups	Measure of parasite load	Data source	Country
Ascaris lumbricoides	49.4	Worm numbers	Thein Hliang et al. (1987) [33]	Myanmar
Ascaris lumbricoides	33.2	Worm numbers	Elkins et al. (1986) [24]	India
Ascaris lumbricoides	27.6	Worm numbers	Croll et al. (1982) [34]	Iran
Trichuris trichiura	34.1	Eggs per gram of faeces	Bundy et al. (1987) [35]	St Lucia
Hookworm (Necator americanus)	15.7	Eggs per gram of faeces	Pullan et al. (2010) [36]	Uganda
Hookworm (Necator americanus)	9.4	Eggs per gram of faeces	Needham et al. (1998) [37]	Vietnam
Schistosoma mansoni	39.7	Eggs per gram of faeces	Kabatereine et al. (2004) [38]	Uganda
Schistosoma mansoni	27.4	Eggs per gram of faeces	Brooker et al. (2007) [39]	Brazil

doi:10.1371/journal.pntd.0002027.t005

- A. Homogeneous model. As defined in equation 7. Treatment is applied to a fraction g of the population with efficacy h and at intervals of τ years.
- B. Heterogeneous model. As described in equations 12 and 13 assuming children and adults are identical epidemiologically. Treatment is applied to a fraction g of school-aged children with efficacy h and at intervals of τ years.
- C. Heterogeneous model with heterogeneous exposure. As described in equations 12 and 13 assuming that children are both twice as strong a source of eggs and have twice the infectious contact rate. Treatment is applied to a fraction g of school-aged children with efficacy h and at intervals of τ years.

It should be noted that under scenario C there will be higher intensity of parasites in children at equilibrium, or baseline, as a consequence of their higher exposure to infection.We use the mathematical model described above (equations 12 and 13) to simulate how the mean intensity of infection would vary for a setting with a given mean intensity and treatment coverage under each of these different scenarios. We run the mathematical model for *A. lumbricoides* and hookworm, which have contrasting transmission

Country	Children enrolled in school, %
Niger	45.5
Eritrea	49.9
Papua New Guinea	50.3
Djibouti	52.1
Chad	60.8
Central African Republic	61.2
Sudan	61.2
Cote d'Ivoire	63.7
Burkina Faso	65.9
Cook Islands	71.3

Table 6. Percentage of female children who are enrolled in school [45].

dynamics, to illustrate the importance of age-dependent mixing in the success of an age-targeted treatment programme.

Results

Fraction of the Total Worm Population in School-Aged Children (5 to 14 Years of Age)

The proportion of worms or egg-output in school-aged children, as calculated using equations 3 and 5 above, is best illustrated by reference to a set of published studies covering the main parasites in different country settings. The first example is that of *A. lumbricoides* in Myanmar from a study by Thein-Hliang et al. [33]. The demography and age intensity data are presented in Figure 3. Demographic surveys for the year of the study reveal that treatment of all children in the 5–14 years age range would expose roughly 49.4% of the worm population to anthelminthics. In terms of the fraction of the parasites killed by a school-based programme assuming a drug efficacy of 95% [5] and school attendance of 95% on the days of treatment gives an overall estimate of 44.5% of the total worm population removed by one round of treatment. Calculations for a study by Elkins et al. [24] of *A. lumbricoides* in



Figure 5. Critical fraction of the population to be treated. The predicted relationship between the critical fraction of the human population to be treated, g_{cr} per annum with efficacy, h, 0.9, and the basic reproductive number, R_0 , and parasite life expectancy, L_1 in years (from equation 11 in the main text). doi:10.1371/journal.pntd.0002027.q005

doi:10.1371/journal.pntd.0002027.t006



Figure 6. Impact of fraction treated on worm burden, prevalence and effective reproduction number. The impact of the fraction of the population treated, g, on A the mean worm burden M, B the prevalence of infection, P and C the effective reproductive number R_e , as described in equation 10. Parameter values are set for A. *lumbricoides* as follows: k = 0.81, $R_0 = 3$, $L_1 = 1$ yr, z = 0.967 and h = 0.95. doi:10.1371/journal.pntd.0002027.g006

Tamil Nadu in India yield a very similar figure for the percentage of worms potentially removed by the treatment of children attending school. Calculations for hookworm (largely \mathcal{N} americanus) in Uganda from a study by Pullan et al. [36] are significantly different, as shown in Figure 4. For this population only 15% of the egg-output is generated by school-aged children. A drug efficacy of 95% and school attendance figure of 85% in a rural region (see Figure 2), produces an estimate of 12% of egg output treated by one round of treatment. These contrasting figures for *A. lumbricoides* and hookworm illustrate the importance of the shape of the age intensity profile to both the efficiency of school-aged targeting and the degree to which such treatment programmes will impact on overall transmission in a population. Analyses of epidemiological studies of Schistosoma mansoni produce figures similar to *A. lumbricoides*, given marked convexity in the intensity by age profiles (age intensity data from Kabatereine et al. [38]). *T. trichiura* is somewhat intermediary between *A. lumbricoides* and hookworm with a degree of convexity but significant worm loads in adults [35,43–44]. The more convex the age intensity profile is, the greater the impact of school-based deworming on overall transmission, provided the peak intensity lies within the age ranges 5-14 years. The continued rise of hookworm intensity in adult age classes yields a low fraction of the total worms in the treated classes.

The calculations presented in Figures 3 and 4 and Table 5 are approximations to the values defined in equations 3 and 5, based on the assumptions that human population size is constant but with an age distribution as documented by the US Bureau of Population and [40], and that the effects of density-dependent fecundity (f_2) are negligible. Polynomial fits are employed to calculate integrals of the functions M(t,a) and p(a) respectively from cross sectional studies of the intensity of infection, and age pyramids (both sexes combined) from national demographic data. The calculations are conservative estimates since they are based on the assumption that all who are eligible attend school. In practice, the percentages recorded in Table 5 must be multiplied by the school attendance data from recent UNESCO studies [45].

The Basic Reproductive Number (R_0) and Parasite Life Expectancy - Their Effects on the Impact of Community-Based Chemotherapy

The impact of the parasite coverage levels presented in Table 5 on transmission under the assumption of homogeneous mixing can be seen using the approximations to the critical treatment coverage, g_c (derived above in relation to the parasite life expectancy, L_1 , and the basic reproduction number, R_0 , as in Figure 5 (equation 11)) and the impact of increasing g on the mean worm load, M, and the prevalence of infection, P (Figure 6, equation 6). Figure 6 reveals that the mean worm load M decays approximately linearly as the fraction treated rises, while prevalence P only begins to fall steeply as the fraction treated approaches the critical value g_c and effective transmission ceases. Note that from Figure 5 the critical fraction treated reaches 1 for high R_0 /low parasite life expectancy. This indicates the parasite may be very difficult to eradicate by treatment with the given efficacy (h = 0.9) when R_0 is large ($\sim >4$, estimated values for particular parasites in Table 3) and L_1 is short (~<1.2 years, values in Table 4) e.g.for A. lumbricoides and T. trichiura in high transmission settings.

Heterogeneity Between Age Classes in Contact with Infective Stages

Further insight on the effect of targeting school-aged children can be gained by considering differential mixing patterns between children and the rest of the population, as outlined above. The results of example simulations are presented in Figures 7 and 8, where the worm burdens in school-aged children and other age groups (where applicable) and averaged across the community are presented for different modelled scenarios, helminths and treatment intervals. The columns of the figures correspond to the scenarios A (homogeneous population), B (homogeneous mixing) and C (heterogeneous mixing). The heterogeneous mixing (scenario C) results in a higher worm burden in the children than in the adults, as is seen in several settings (note this model does not include any immunity). All models have the same mid-range R_0 value of 3. Treatment in the homogeneous model is made comparable with the heterogeneous model by setting coverage to gn_c . We have simulated these scenarios for A. lumbricoides, with a life expectancy of 1 year (Figure 7) and hookworm with a life expectancy of 2.5 years (Figure 8).



Figure 7. Effect of regular treatment on mean *A. lumbricoides* **worm burden for different models.** A homogeneous population (left column), B heterogeneous population with uniform transmission dynamics (central column) and C heterogeneous population with greater contribution from children (right column) as in the text. The two rows represent annual and half-yearly treatment respectively. For all runs, basic reproduction number is 3 and worm lifespan is 1 year. Other parameters (as defined for equations 6 and 7): $\mu_2 = 5/\text{yr}$, k = 0.7, z = 0.93. doi:10.1371/journal.pntd.0002027.q007

The most striking feature of Figure 7 and Figure 8 is the very modest impact of treatment of children on transmission, even at high levels of efficacy (95%) and coverage (85%). The treated children do have large benefits in terms of periods free of worms or with low worm burdens. However, the effect of treating children on worm burdens in the larger community is small. This reflects the proportion of the worm population actually reached by treatment, even though the chosen value (30%) is at the high end of school-attending fraction of the population (Table 5). Decreasing the treatment interval (bottom row in each figure) has only a moderate effect. The two group equal mixing model (scenario B) shows the direct effect of school-based treatment on school-aged children and also the indirect effect on adults through the

reduction in infectious material in the community. The rate of bounce-back after treatment is slightly reduced in the heterogeneous model as compared to the homogeneous one (scenario B versus scenario A). This means that homogeneous descriptions (e.g. the homogeneous model, A) of non-uniform treatment regimens (targeted at some portion of the population) will always underestimate the time to recover to pre-treatment levels.

Scenario C mimics what we believe to be a more realistic epidemiological scenario, with school-aged children contributing twice as much infectious material as other age groups and also being twice as exposed, resulting in higher worm burdens in school-aged children. This is more likely to be observed for *A. lumbricoides* (e.g. Figure 3) than for hookworm (e.g. Figure 4). As



Figure 8. Effect of regular treatment on mean worm burden of hookworm for different models. As in Figure 7, the columns are A homogeneous model, B heterogeneous population with uniform transmission dynamics and C heterogeneous population with greater contribution from children, as in the text and different treatment intervals (rows). Simulations for basic reproduction number, $R_0 = 3$, and worm lifespan is 2.5 years. Other parameters as in Figure 7. The two rows represent two-yearly and yearly treatment respectively. doi:10.1371/journal.pntd.0002027.g008

such, the effect of treatment on the school-aged group is quite pronounced, but the impact at the community level, sometimes termed the 'herd impact' is only marginally improved, due again to the small proportion of worms treated. These simulations highlight the importance of mixing patterns in determining the effectiveness of school-based treatment programmes.

Discussion

School-based approaches to deworming children have many advantages in terms of ease of access in urban and rural regions and the ability to link with other nutritional, health and education initiatives in order to try and minimize delivery and logistic costs. Advocacy for this approach to the control of STHs and the morbidity they induce has been made by many over the past decade [1,30,46]. However, with increased drug donations to support such programmes, it is now crucial to evaluate the benefits and disadvantages of such an approach. The limitations of this approach has already been implicitly acknowledged in the WHO recommendations to additionally target pre-school children, women of child-bearing age and high risk adults where possible [4] and previous identification of adults as a possible reservoir of infection [47]. However, there are many countries where only school-based deworming is currently under consideration [11]. Of particular importance in this context is the impact of school-based treatment on transmission of the parasites in the entire commu-



Figure 9. Schematic illustration of the impact of school-based deworming on the transmission of parasites. The number of parasites affected by a school-based deworming programme is never 100%, it is reduced by the efficacy of the drug, the proportion of the population of school age and their intensity of infection, as well as school enrolment and attendance on a deworming day. The impact of such a treatment programme on transmission is less easily quantified and depends on the details of transmission between different age-groups in the population. For further details, see main text. doi:10.1371/journal.pntd.0002027.g009

nity, including the pre-school and adult age groups. In particular, sustained transmission (and thus production of infective stages) in other age groups will influence the frequency of treatment in school-aged children required to sustain infection at very low levels.

A limited number of field-based studies of mass chemotherapy have suggested that adult age groups who do not have access to treatment still benefit from school-based deworming as a result of its impact on the overall intensity of transmission within the population [30,33,48]. The treatment of a few (heavily infected individuals) can impact on the effective reproductive number and therefore reduced exposure to infective stages in those untreated. This effect is analogous to the concept of herd immunity in community-based vaccination programmes where vaccination reduces transmission to those still susceptible to infection. In the context of worms and chemotherapy, the herd impact arises from the reduction in the output of infective stages in faeces that result in the contamination of the environment in which the community of all age groups lives. A similar indirect protection among untreated individuals is the impact upon older children and adults seen after age-targeted (1-10 year olds) large-scale treatment of azithromycin against trachoma [49].

We employed two approaches to examine the impact of school deworming programmes on infection within the entire community, the 'herd impact' of the treatment programme. The first was empirical and based on the calculation of the proportions of the population in the school attendance age groups, the fraction of these age groups who attend school and the fraction of the total worm population harboured by in these school-aged children. The results of a set of calculations are summarised in Table 5. It should be noted that the age-profile and population pyramids used here were not exactly matched, and therefore the calculations will not be precise for a particular point in time. It highlights the need for more recent data for these pathogens, as highlighted in recent articles [50–51]. In many of the low-income settings in which STHs are highly prevalent, the population pyramid is approximately exponential in shape, meaning that from 20–30% (Table 2, Figure 3A and Figure 4A) are of school age. However, for some STHs, such as A. lumbricoides, the highest intensity infections are in this age-group (Figure 3B) and therefore a large proportion of worms are targeted by school-aged treatment (up to 50% in our examples, Table 5). These predictions are similar to empirical findings by Bundy et al. [30] who evaluated the impact of agetarget chemotherapy on community transmission on the island of Montserrat, West Indies, where T. trichiura, which has similar transmission dynamics to A. lumbricoides, was the predominant species and had a prevalence of 12% (A. lumbricoides and hookworm occurred at < 2%). The authors found that 4-monthy treatment of 2-15 year olds had a subsidiary effect on intensity of untreated 16-25 year olds. In contrast to A. lumbricoides and T. trichiura which exhibit age-convexity in intensity, hookworm intensity tends to monotonically increase with age (e.g. Figure 4B), as has been seen in several studies [52-54], and therefore a smaller proportion of worms or egg output are targeted by school-aged treatment (<10% in one of our examples, Table 5).

The different age profiles for these helminth species are a result of differing behaviour patterns, force of infection, heterogeneous exposure and, arguably, immunity and genetic pre-disposition. The details of the biology which generate these patterns do not need to be understood for the calculation of the proportion of worms treated. However, to estimate the impact of treating children on transmission in the larger community we need a combination of additional studies and novel modelling analyses. Our simulations (Figures 7 and 8) show the importance of understanding the nature of the interaction between school-aged children and the rest of the community in order to optimise treatment programmes in the longer term. It is possible that school-aged deworming will give little benefit outside the children being treated and that intensifying either the treatment coverage within this age-group and/or increasing the frequency of treatment (bottom rows in Figures 7 and 8) may not lead to the desired benefits in the larger community. In this case, other strategies will be needed to increase the potential for long-term control and, if the breakpoint can be crossed by combinations of interventions, eventual elimination of these helminths.

As the schematic, Figure 9, shows, the impact of school-based treatment programmes on transmission in the larger community is diluted by a number of effects. The benefits of deworming for the affected children are many, but if we are to plan for long-term control and, in the longer term, elimination of these pathogens we need to consider strategies that will reduce transmission from year to year, as is already being discussed in some settings [55], particularly for schistosomiasis [56]. We also require an understanding of the species mix in each setting so as to tailor the design of interventions according to the underlying transmission dynamics. As this paper shows, there are many outstanding data gaps and needs for new modelling studies both to understand the dynamics of transmission under such programmes, and to design optimal treatment strategies for the future.

Acknowledgments

The authors thank the anonymous reviewers and editors for their helpful comments.

Author Contributions

Conceived and designed the experiments: RMA JET TDH. Performed the experiments: RMA JET TDH. Analyzed the data: RMA JET TDH. Contributed reagents/materials/analysis tools: RMA JET TDH. Wrote the paper: RMA JET RLP SJB TDH.

References

- WHO (2012) Accelerating work to overcome the global impact of neglected tropical diseases - A roadmap for implementation. Geneva: World Health Organization.
- WHO (2011) Soil-transmitted helminthiases: estimates of the number of children needing preventive chemotherapy and number treated, 2009. Wkly Epidemiol Rec 86: 257–266.
- Pullan RL, Brooker SJ (2012) The global limits and population at risk of soiltransmitted helminth infections in 2010. Parasit Vectors 5: 81.
- WHO (2006) Preventive Chemotheraphy in Human Helminthiasis. Geneva, Switzerland: World Health Organization.
- Keiser J, Utzinger J (2008) Efficacy of current drugs against soil-transmitted helminth infections: systematic review and meta-analysis. JAMA 299: 1937– 1948.
- Jia TW, Melville S, Utzinger J, King CH, Zhou XN (2012) Soil-transmitted helminth reinfection after drug treatment: a systematic review and meta-analysis. PLoS Negl Trop Dis 6: e1621.
- Allen JE, Maizels RM (2011) Diversity and dialogue in immunity to helminths. Nat Rev Immunol 11: 375–388.
- Bourke CD, Maizels RM, Mutapi F (2011) Acquired immune heterogeneity and its sources in human helminth infection. Parasitology 138: 139–159.
- Ziegelbauer K, Speich B, Mäusezahl D, Bos R, Keiser J, et al. (2012) Effect of sanitation on soil-transmitted helminth infection: systematic review and metaanalysis. PLoS Med 9: e1001162.
- Anderson RM, May RM (1985) Helminth infections of humans mathematical models, population dynamics, and control. Adv Parasitol 24: 1–101.
- WHO (2011) Helminth control in school-age children. Geneva, Switzerland: World Health Organization.
- Brooker S, Hotez PJ, Bundy DAP (2010) The global atlas of helminth infection: mapping the way forward in neglected tropical disease control. PLoS Negl Trop Dis 4: e779.
- Lustigman S, Prichard RK, Gazzinelli A, Grant WN, Boatin BA, et al. (2012) A research agenda for helminth diseases of humans: the problem of helminthiases. PLoS Negl Trop Dis 6: e1582.
- Brooker S, Kabatereine NB, Fleming F, Devlin N (2008) Cost and costeffectiveness of nationwide school-based helminth control in Uganda: intracountry variation and effects of scaling-up. Health Policy Plan 23: 24–35.
- Leslie J, Garba A, Oliva EB, Barkire A, Tinni AA, et al. (2011) Schistosomiasis and soil-transmitted helminth control in Niger: cost effectiveness of school based and community distributed mass drug administration [corrected]. PLoS Negl Trop Dis 5: e1326.
- Ottesen EA, Hooper PJ, Bradley M, Biswas G (2008) The global programme to eliminate lymphatic filariasis: health impact after 8 years. PLoS Negl Trop Dis 2: e317.
- 17. Anonymous (2012) London declaration on neglected tropical diseases.
- Anderson RM, May RM (1991) Infectious Diseases of Humans: Dynamics and Control. Oxford, UK: Oxford University Press.
- Michiels M, Hendriks R, Heykants J, van den Bossche H (1982) The pharmacokinetics of mebendazole and flubendazole in animals and man. Arch Int Pharmacodyn Ther 256: 180–191.
- Marriner SE, Morris DL, Dickson B, Bogan JA (1986) Pharmacokinetics of albendazole in man. Eur J Clin Pharmacol 30: 705–708.
- Levecke B, Mekonnen Z, Albonico M, Vercruysse J (2012) The impact of baseline faecal egg counts on the efficacy of single-dose albendazole against Trichuris trichiura. Trans R Soc Trop Med Hyg 106: 128–130.
- Vercruysse J, Behnke JM, Albonico M, Ame SM, Angebault C, et al. (2011) Assessment of the anthelmintic efficacy of albendazole in school children in seven countries where soil-transmitted helminths are endemic. PLoS Negl Trop Dis 5: e948.
- 23. Jordan P, Webbe G (1969) Human Schistosomiasis. London: William Heinemann Medical Books Ltd.
- Elkins DB, Haswell-Elkins M, Anderson RM (1986) The epidemiology and control of intestinal helminths in the Pulicat Lake region of Southern India. I. Study design and pre- and post-treatment observations on Ascaris lumbricoides infection. Trans Roy Soc Trop Med Hyg 80: 774–792.
- Holland CV, Asaolu SO, Crompton DWP, Stoddart RC, Macdonald R, et al. (1989) The epidemiology of Ascaris lumbricoides and other soil-transmitted helminths in primary school children from Ile-Ife, Nigeria. Parasitology 99 Pt 2: 275–285.
- Hall A, Anwar KS, Tomkins A, Rahman L (1999) The distribution of Ascaris lumbricoides in human hosts: a study of 1765 people in Bangladesh. Trans R Soc Trop Med Hyg 93: 503–510.
- Anderson RM, May RM (1985) Helminth Infections of Humans Mathematical-Models, Population-Dynamics, and Control. Advances in Parasitology 24: 1–101.
- Thein H, Than S, Myat Lay K (1990) Control of ascariasis through age-targeted chemotherapy: impact of 6-monthly chemotherapeutic regimens. Bull World Health Organ 68: 747–753.
- Anscombe FJ (1950) Sampling theory of the negative binomial and logarithmic series distributions. Biometrika 37: 358–382.

- Bundy DAP, Wong MS, Lewis LL, Horton J (1990) Control of geohelminths by delivery of targeted chemotherapy through schools. Trans R Soc Trop Med Hyg 84: 115–120.
- Hurlimann E, Schur N, Boutsika K, Stensgaard AS, Laserna de Himpsl M, et al. (2011) Toward an open-access global database for mapping, control, and surveillance of neglected tropical diseases. PLoS Negl Trop Dis 5: e1404.
- Prichard RK, Basanez MG, Boatin BA, McCarthy JS, Garcia HH, et al. (2012) A research agenda for helminth diseases of humans: intervention for control and elimination. PLoS Negl Trop Dis 6: e1549.
- Hlaing T, Saw T, Lwin M (1987) Reinfection of people with Ascaris lumbricoides following single, 6-month and 12-month interval mass chemotherapy in Okpo village, rural Burma. Trans R Soc Trop Med Hyg 81: 140– 146.
- Croll NA, Anderson RM, Gyorkos TW, Ghadirian E (1982) The population biology and control of Ascaris lumbricoides in a rural community in Iran. Trans R Soc Trop Med Hyg 76: 187–197.
- Bundy DAP, Cooper ES, Thompson DE, Didier JM, Simmons I (1987) Epidemiology and population dynamics of Ascaris lumbricoides and Trichuris trichiura infection in the same community. Trans R Soc Trop Med Hyg 81: 987–993.
- Pullan RL, Kabatereine NB, Quinnell RJ, Brooker S (2010) Spatial and genetic epidemiology of hookworm in a rural community in Uganda. PLoS Negl Trop Dis 4: e713.
- Needham C, Kim HT, Hoa NV, Cong LD, Michael E, et al. (1998) Epidemiology of soil-transmitted nematode infections in Ha Nam province, Vietnam. Trop Med Int Health 3: 904–912.
- Kabatereine NB, Brooker S, Tukahebwa EM, Kazibwe F, Onapa AW (2004) Epidemiology and geography of Schistosoma mansoni in Uganda: implications for planning control. Trop Med Int Health 9: 372–380.
- Brooker S, Alexander N, Geiger S, Moyeed RA, Stander J, et al. (2006) Contrasting patterns in the small-scale heterogeneity of human helminth infections in urban and rural environments in Brazil. Int J Parasitol 36: 1143– 1151.
- United States Census Bureau (2012) International Data Base. Available: Accessed 19th May 2012.
- UNICEF (2012) Child Info Statistical Tables, Education. Available: http:// www.childinfo.org/education_1059.htm. Accessed 19th May 2012.
- Chan MS, Guyatt HL, Bundy DAP, Medley GF (1994) The development and validation of an age-structured model for the evaluation of disease control strategies for intestinal helminths. Parasitology 109: 389–396.
- 43. Unhanand M, Ngamwat W, Permpipat U (1983) Study on the toxicity of maklua (Diospyros mollis (Griff)) used as a traditional anthelmintic in Thailand. Collected Papers on the control of Soi-Transmitted Helminthaises by the APCO Research Group 1980–1982. pp. 161–179.
- Bradley M, Chandiwana SK, Bundy DAP, Medley GF (1992) The epidemiology and population biology of Necator americanus infection in a rural community in Zimbabwe. Trans R Soc Trop Med Hyg 86: 73–76.
- UNESCO (2012) Education Data, UNESCO Institute of Statistics. Available: http://www.uis.unesco.org/Education/. Accessed 20th May 2012.
- Baird S, Hicks JH, Kremer M, Miguel E (2011) Worms at Work: Long-run Impacts of Child Health Gains. Available: http://www.povertyactionlab.org/ publication/worms-work-long-run-impacts-child-health-gains. Accessed 19 May 2012.
- 47. Njenga SM, Mwandawiro CS, Muniu E, Mwanje MT, Haji FM, et al. (2011) Adult population as potential reservoir of NTD infections in rural villages of Kwale district, Coastal Kenya: implications for preventive chemotherapy interventions policy. Parasit Vectors 4: 175.
- Asaolu SO, Holland CV, Crompton DWP (1991) Community control of Ascaris lumbricoides in rural Oyo State, Nigeria: mass, targeted and selective treatment with levamisole. Parasitology 103: 291–298.
- House JI, Ayele B, Porco TC, Zhou Z, Hong KC, et al. (2009) Assessment of herd protection against trachoma due to repeated mass antibiotic distributions: a cluster-randomised trial. Lancet 373: 1111–1118.
- Basáñez MG, McCarthy JS, French MD, Yang GJ, Walker M, et al. (2012) A research agenda for helminth diseases of humans: modelling for control and elimination. PLoS Negl Trop Dis 6: e1548.
- Anderson R, Hollingsworth TD, Truscott J, Brooker S (2012) Optimisation of mass chemotherapy to control soil-transmitted helminth infection. Lancet 379: 289–290.
- Sturrock RF (1966) Hookworm studies in Uganda: investigations at Teboke in Lango district. East Afr Med J 43: 430–438.
- Sturrock RF (1966) Hookworm studies in Tanganyika (Tanzania): investigations at Hombolo in the Dodoma region. East Afr Med J 43: 315–322.
 Chunge RN, Karumba PN, Andala EO (1986) Hookworm species in patients
- Chunge RN, Karumba PN, Andala EO (1986) Hookworm species in patients from Kenyatta National Hospital Nairobi. Ann Trop Med Parasitol 80: 147– 148.
- Knopp S, Stothard JR, Rollinson D, Mohammed KA, Khamis IS, et al. (2011) From morbidity control to transmission control: time to change tactics against helminths on Unguja Island, Zanzibar. Acta Trop.
- Rollinson D, Knopp S, Levitz S, Stothard JR, Tchuente LA, et al. (2012) Time to set the agenda for schistosomiasis elimination. Acta Trop.

- Thein H, Than S, Htay Htay A, Myint L, Thein Maung M (1984) Epidemiology and transmission dynamics of Ascaris lumbricoides in Okpo village, rural Burma. Trans R Soc Trop Med Hyg 78: 497–504.
- Martin J, Keymer A, Isherwood RJ, Wainwright SM (1983) The prevalence and intensity of Ascaris lumbricoides infections in Moslem children from northern Bangladesh. Trans R Soc Trop Med Hyg 77: 702–706.
- Chai JY, Kim KS, Hong ST, Lee SH, Seo BS (1985) Prevalence, worm burden and other epidemiological parameters of Ascaris lumbricoides infection in rural communities in Korea. Kisaengchunghak Chapchi 23: 241–246.
- Bradley M, Chandiwana SK, Bundy DAP (1993) The epidemiology and control of hookworm infection in the Burma Valley area of Zimbabwe. Trans R Soc Trop Med Hyg 87: 145–147.
- Quinnell RJ, Slater AF, Tighe P, Walsh EA, Keymer AE, et al. (1993) Reinfection with hookworm after chemotherapy in Papua New Guinea. Parasitology 106: 379–385.